Author's response to reviews

Title: Breathlessness is associated with urinary incontinence in men: A community-based study

Authors:

Fumi Hirayama (F.Hirayama@curtin.edu.au)
Andy H Lee (Andy.Lee@curtin.edu.au)
Tetsuo Hiramatsu (t.hiramatsu@komakihp.gr.jp)
Yoshimasa Tanikawa (tanikawa@hm5.aitai.ne.jp)

Version: 4 Date: 22 December 2009

Author's response to reviews:

Response to Reviewer 3

1. Abstract: “risk” changed to “odds” as suggested.

2. Methods: Information has been added regarding the 5-level categorical physical activity variable being dichotomized to ‘not active’ and ‘active’.

3. Statistical analysis: the presence of co-morbidities has been added in the multivariate model as suggested.

4. Tables: abbreviations are now defined.